Math 138A Winter 2006 R. Schultz

UPDATED GENERAL INFORMATION — JANUARY 25, 2006

This is a revised quiz and examination schedule. One further change is that the midterm examinations will be take-home assignments rather than in-class tests.

QUIZ SCHEDULE: Tuesday, February 7, and Tuesday, February 28.

TAKE-HOME EXAMINATION SCHEDULE: These will be due at the beginning of class on Wednesday, February 15, and Wednesday, March 8, and they will be posted at least one week before they are due.

Previous statements about the final examination are not affected by these changes.